Author's response to reviews

Title: Fructose-1, 6-diphosphate (FDP) as a novel antidote for yellow oleander-induced cardiac toxicity: A randomized controlled double blind study

Authors:

Indika B Gawarammana (indika@sactrc.org)
Fahim Mohamed (fahim@sactrc.org)
Steven J Bowe (Steven.Bowe@newcastle.edu.au)
G A Rathnathilake (contact@sactrc.org)
K Narangoda (enquiry@sactrc.org)
Andrew H Dawson (ahdawson@gmail.com)
Nick A Buckley (n.buckley@unsw.edu.au)
Shifa Azher (awanthi@sactrc.org)

Version: 2 Date: 15 April 2010

Author's response to reviews: see over
Date: 11th April 2010

Ref: MS: 3660347193200204

Title: Fructose-1, 6-diphosphate (FDP) as a novel antidote for yellow oleander-induced cardiac toxicity: A randomized controlled double blind study

The Editor
BMC-Series Journals
BioMed Central

Dear editor,

We appreciate the feedback provided to the study protocol “Fructose-1, 6-diphosphate (FDP) as a novel antidote for yellow oleander-induced cardiac toxicity: A randomized controlled double blind study” and are pleased that the reviewers found the study relevant and important. We have gone through the comments and incorporated, where possible, the feedback from the reviewer into the paper. In addition to responding to reviewer’s comments, we have included flow a chart as suggested by the editorial adviser and also ensured that the study protocol is written in accordance with the CONSORT guideline. We also take this opportunity to add new author to this manuscript. A version with tracked changes is attached.

Yours sincerely

Fahim Mohamed
Feedback to reviewer’s comments

I would like to see further clarification of the agent to be used in the rescue bolus. Is this the same agent as the first bolus, a cross over or ?

Yes, the rescue bolus is the same agent as first bolus to which the patient was randomised.

I would like to see a clearer statement regarding intention to treat analysis in the body of the manuscript.

We have elaborated briefly about intention to treat analysis on the manuscript according to reviewer’s suggestion. As a result, we have added a new reference.